Author's response to reviews

Title: Global Health Education in Germany: needs, deficits and potential benefits - a cross-sectional study among medical students (Part 1 of 2: Mobility patterns & educational needs and demands)

Authors:

   Kayvan Bozorgmehr (kayvan.bozorgmehr@googlemail.com)
   Kirsten Schubert (ki.schubert@googlemail.com)
   Johannes Menzel-Severing (menzel-severing@web.de)
   Peter Tinnemann (Peter.Tinnemann@charite.de)

Version: 2 Date: 3 March 2010

Author's response to reviews: see over
Cover Letter to the revised manuscript MS: 1006277917325048

‘Global Health Education in Germany: needs, deficits and potential benefits - a cross-sectional study among medical students (Part 1 of 2: Mobility patterns & educational needs and demands)’

This manuscript is the revised submission of a two-part series on global health education in Germany. The major changes done on Part 1 of the series during the revision are the following:

- We have added additional findings to the results section of the previously submitted version regarding
  - the supply of global health education at respondents’ medical faculties and
  - the demand for education in global health among our sample.

As a consequence, these amendments went along with according changes in other sections as well, i.e. in the abstract, background, methods and discussion section.

The second part of the series builds on the findings presented in Part 1 and is concerned with medical students’ knowledge of global health issues and potential benefits of global health education.

For further questions regarding the revision of this manuscript, please do not hesitate to contact me.

Yours sincerely,

Kayvan Bozorgmehr